

G OPEN ACCESS

Citation: Carcauzon V, Herrera JP, Kaufman K, Baudino F, Wickenkamp N, Randriamoria TM, et al. (2024) Astroviruses in terrestrial Malagasy mammals. PLoS Negl Trop Dis 18(6): e0012263. https://doi.org/10.1371/journal.pntd.0012263

Editor: Abdallah M. Samy, Faculty of Science, Ain Shams University (ASU), EGYPT

Received: May 5, 2023

Accepted: June 1, 2024

Published: June 14, 2024

Copyright: © 2024 Carcauzon et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All sequences are made available in Genbank: ON652986-ON653001 and OP727997-OP728001.

Funding: Funding was provided by the joint National Institutes of Health - National Science Foundation - National Institute of Food and Agriculture "Ecology and Evolution of Infectious Disease" award (R01-TW011493 to CN) and a Duke University Provost's Collaboratory grant (to CN), the Duke University Bass Connections program (to CN), and by a grant from the Grainger Bioinformatics Center at the Field Museum of **RESEARCH ARTICLE**

Astroviruses in terrestrial Malagasy mammals

Victoria Carcauzon¹, James P. Herrera^{2,3}, Kayla Kaufman^{2,4}, Fiona Baudino¹, Natalie Wickenkamp², Toky M. Randriamoria⁵, Voahangy Soarimalala⁵, Steven M. Goodman^{5,6}, Charles L. Nunn^{2,7‡*}, Camille Lebarbenchon^{1†}, Pablo Tortosa^{1†}

 Université de La Réunion, Unité Mixte de Recherche Processus Infectieux en Milieu Insulaire Tropical (UMR PIMIT), CNRS 9192, INSERM 1187, IRD 249, Plateforme Technologique CYROI, Sainte Clotilde, La Réunion, France, 2 Evolutionary Anthropology, Duke University, Durham, North Carolina, United States of America, 3 Duke Lemur Center SAVA Conservation, Durham, North Carolina, United States of America, 4 University of California Santa Barbara, Department of Ecology, Evolution, and Marine Biology, University of California, Santa Barbara, California, United States of America, 5 Association Vahatra, Antananarivo, Madagascar, 6 Field Museum of Natural History, Chicago, Illinois, United States of America, 7 Duke Global Health Institute, Durham, North Carolina, United States of America

‡ These authors are co-last authors on this work.
* clnunn@duke.edu

Abstract

Small terrestrial mammals are major hosts of infectious agents responsible for zoonotic diseases. Astroviruses (AstVs)-the cause of non-bacterial gastroenteritis mainly affecting young children-have been detected in a wide array of mammalian and avian host species. However, understanding the factors that influence AstV infection within and across hosts is limited. Here, we investigated the impact of land use changes on AstVs in terrestrial small mammals in rural northeastern Madagascar. We sampled 515 small mammals, representing seven endemic and four introduced species. Twenty-two positive samples were identified, all but one of which were found in the introduced species Mus musculus and Rattus rattus (family Muridae), with a positivity rate of 7.7% (6/78) and 5.6% (15/266), respectively. The non-introduced rodent case was from an endemic shrew-tenrec (family Tenrecidae). We found the highest positivity rate of AstVs infection in brushy regrowth (17.5%, 7/40) as compared to flooded rice fields (4.60%, 8/174), secondary forest (4.1%, 3/74), agroforest (3.6%, 1/28), village (2.61%, 3/115), and semi-intact forest (0%, 0/84). A phylogenetic analysis revealed an association between AstVs and their rodent host species. None of the viruses were phylogenetically related to AstVs previously described in Malagasy bats. This study supports AstV circulation in synanthropic animals in agricultural habitats of Madagascar and highlights the need to assess the spillover risk to human populations in rural areas.

Author summary

This article describes the diversity of Astroviruses (AstVs) shed by Malagasy terrestrial small mammals and explores the impact of habitat degradation on positivity rate in animal reservoirs. Using a sample of 515 terrestrial small mammals from northeastern Madagascar, we show that AstV infection is almost exclusively restricted to rats and mice, suggesting that AstVs from terrestrial mammals were introduced to Madagascar. A Natural History (to SMG and PT). In addition, the collaboration between Duke University (JPH) and University of La Réunion (PT) was supported by the Transatlantic Research Partnership, a program of the FACE Foundation (https://face-foundation.org/) and French Embassy. CN, KK and VS had all or part of their salary provided by NIH R01-TW011493. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

phylogenetic analysis including AstVs sequences from Malagasy small mammals reveals a strong structuring of AstV sequences within host species, indicating limited host switching of these viruses among small mammals. Finally, positivity rates show that some habitat types are particularly conducive to AstV infection in reservoirs, such as rice fields where mice exhibit maximal infection. Altogether, presented data encourage the need for assessing the actual exposure of the local human population to the AstVs in small mammals, especially rice-field farmers.

Introduction

Astroviruses (AstVs) are single-stranded non-enveloped RNA viruses belonging to the family *Astroviridae*. They are responsible worldwide for 2–8% of non-bacterial gastroenteritis affecting children under five years of age [1], making them the third most common viral agent of acute diarrhea after rota- and noro-viruses [2–4]. AstVs are mainly transmitted via the fecaloral route or indirectly through consumption of contaminated food or water. Because these viruses can persist in contaminated environments for long periods, non-contact transmission is an important mechanism of spread [5]. Symptoms of AstV-caused gastroenteritis include diarrhea, which can lead to severe disease in immunocompromised children or adults [6]. Recently, other types of AstVs (e.g. HAstV-MLB and HAstV-VA/HMO) have been identified in humans and are thought to be associated with central nervous system infections [7].

AstVs have been detected in more than 80 non-human host species including a wide diversity of small terrestrial mammals (e.g., rodents and shrews) with a nearly global distribution that includes North America [8], Asia [9,10], and Africa [11]. AstVs are also common in bats [12], and have been recently detected in a range of species in the western Indian Ocean, specifically Mozambique [13], Madagascar [14], and Reunion Island [15].

Although no information is available regarding AstV in human populations from Madagascar, enteric diseases are highly prevalent in the country, and these viruses might be of medical concern. Since small mammals, and especially introduced rodents, are commensal to human populations, we investigated AstV circulation in introduced and endemic small terrestrial mammals in a humid forest area of northeastern Madagascar, focusing on viral positivity rate, diversity, and effect of land use and season on viral infection. The study was conducted within and adjacent to Marojejy National Park over three years (2017–2019). The protected area has an exceptional level of vertebrate diversity [16] and the surrounding areas are characterized by different forms of agricultural use. These landscapes are used by local people for a variety of economic and subsistence activities, including rice, vanilla and other agricultural products, harvesting of trees for cooking fuel and house construction, and limited livestock pasture land [17]. We sampled animals in plots along a land use gradient encompassing villages, agricultural zones, secondary forest fragments, and disturbed native forest within the national park.

We hypothesized that habitat disturbance influences AstV transmission in terrestrial small mammals by affecting the community structure of mammal reservoirs and by altering infection positivity rates in reservoirs, such as rice farming activities that may increase viral positivity rate due to non-contact transmission in humid environments. A previous study on bats found an effect of season but no effect of land use on AstV infection prevalence in bats [18]. However, terrestrial small mammals have more limited dispersal capacities than bats, and are in permanent and direct contact with the ground. As a result, it is likely that spatio-temporal dynamics of AstVs shed by bats and terrestrial small mammals are distinct.

Methods

Ethics statement

The protocols of animal capture and handling, and collection of biological material were approved by the Institutional Animal Care and Use Committee of Duke University (protocol numbers A002-17-01 & A262-19-12) and by Malagasy authorities (permit numbers 289/17 & 146/18—MEEF/SG/DGF/DSAP/SCB).

Study sites and sampling seasons

This study was carried out in northeastern Madagascar, in the SAVA Region, within and adjacent to Marojejy National Park, in the Manantenina River valley and based at a campsite (-14.4662, 49.7986), 2 km from the village of Mandena. This region of the island is well-known for the cultivation of vanilla. Forested areas that have been transformed into a form of agroforestry are composed of largely non-native tree plantations or fields suitable for vanilla and rice cultivation. These modified habitats still hold some endemic species of small mammals [17,19]. Small mammals were trapped across six habitat types encompassing a range of forest cover and human use (See Fig 1). Forest habitat types included semi-intact forest and secondary forest of the lowland moist evergreen forest formation, as defined by Gautier et al. (2018) [20]. The other non-forested habitat types were crop fields and included mixed agriculture, paddy rice, vanilla, and savoka (regenerating secondary vegetation after forest clearing, and generally composed of non-native and invasive plants). In many cases, the initial incentive for forest clearing was to establish areas of shifting or swidden agriculture (tavy in Malagasy). This type of agriculture depletes the soil of nutrients, only remains productive for a few seasons then the site is abandoned, and subsequently develops to dense brushy regrowth or *savoka*. Each habitat type was sampled during two different seasons: dry season (June-August 2018, Mandena and June-July, Manantenina) and wet season (Novembre-Decembre 2017, Mandena).



Fig 1. Astrovirus detection in small terrestrial Malagasy mammals captured in the different habitat types in and around the Marojejy National Park. (A) Map presenting the different habitat types at the study sites and (B) map showing the locations of each trapping grid together with the exact position of trapped animals colored in green (AstV negative) or red (AstV positive). The map was obtained using leaflet R package and the Map image is the intellectual property of Esri and is used herein under license. Source: Esri, i-cubed, USDA, USGS, AEX, Getmapping, Aerogrid, IGN, IGP, UPR-EGP, and the GIS User Community (https://arcg.is/1uyPfH).

https://doi.org/10.1371/journal.pntd.0012263.g001

Animal capture and handling

Small mammals were trapped as previously described [17] and included introduced Muridae rodents and Soricidae shrews, as well as endemic rodents (subfamily Nesomyinae) and tenrecs (family Tenrecidae). All animals were identified in the field based on a range of external characters [21] and for some collected individuals based on cranio-dental features. The study included 515 animal samples, of which 375 were trapped in 2017/2018 and were previously screened for pathogenic Leptospira [17] while 140 were sampled in 2019 following the same exact trapping methods. Euthanasia was performed by cervical dislocation except for the larger tenrecs (Setifer setosus and Tenrec ecaudatus) for which a lethal dose of xylazine was administered via inter-muscular injection. For euthanized mammals (all introduced animals, as well as a subset of native animals, typically less than 10 per site), kidney, bladder, lung, colon, urine, blood, and ectoparasites were collected and immediately stored in 75% ethanol. Samples were stored at room temperature in the field and then brought to the laboratory where they were stored at -80°C. All mammal specimens were deposited in the Université d'Antananarivo, Mention Zoologie et Biodiversité Animale, and formerly the Département de Biologie (UADBA). Final species identification of the specimens was made by small mammal specialists based on external and, in some cases, cranio-dental morphological characters.

Nucleic acids extraction, PCR diagnosis, and sequencing

We used colon tissue samples for the screening of AstroV. Since tissues were stored in EtOH70%, colon samples were rehydrated for 24 hours with osmosis water at 4°C before nucleic acid extraction. Following rehydration for each sample, approximately 20 mg of tissue was placed in 200 µL of ATL buffer (QIAGEN, Valencia, California, USA) containing 20 µL of Proteinase K. These samples were incubated at 56°C for 6 hours to achieve complete lysis and then centrifuged for 15 min at 16 000 rpm. The supernatant was added to 180 µL of VXL buffer (QIAGEN, Valencia, California, USA) and nucleic acid extraction was performed using the Cador Pathogen kit and the QIAcube XT automated system (QIAGEN, Valencia, California, USA). Reverse transcription was performed from 10 µL of nucleic acids using ProtoScript II Reverse Transcriptase and Random Primer 6 (New England BioLabs, Ipswich, Massachusetts, USA) under the following thermal conditions: 70°C for 5 min, 25°C for 10 min, 42°C for 50 min, and 65°C for 20 min, with an Applied Biosystems 2720 thermal cycler (Thermo Fisher Scientific, Waltham, Massachusetts, USA) [14]. Produced cDNAs were tested for the presence of the AstVs RNA-dependent RNA polymerase (RdRp) using a semi-nested PCR [22] with the GoTaq G2 Hot Start Green Master Mix (Promega, Madison, Wisconsin, USA). After electrophoresis on 2.0% agarose gel stained with 2% GelRed (Biotium, Hayward, California, USA), PCR products of the expected size were gel purified with the QIAquick purification kit (QIA-GEN, Valencia, California, USA). Sanger sequencing was performed on both strands (Genoscreen, Lille, France) and only those samples for which we obtained an AstV sequence were considered positive in the analyses.

Statistical analyses

We compared the positivity rates among habitat types and species using Fisher's exact tests. To investigate which factors predicted AstV infection in animals, we performed model selection based on the corrected Akaike information criterion (AICc) using the dredge function in the MuMIn package in R (package version 1.47.1, Barton 2022). All possible additive models containing any or all of the set of predictors: habitat type (secondary forest, agroforest, brushy regrowth, flooded rice, village), season (end of dry, dry), and the village (Mandena, Manantenina) in which the animal was captured, and the species (*Rattus rattus, Mus musculus*), sex

(male, female), and mass (g) of the individual. A total of 64 models were considered. We retained all models in which a cumulative sum of Akaike weights less than or equal to 95% to estimate averaged model coefficients and parameter importance (sum of weights across all models in which the variable appears).

Phylogenetic analysis

All RdRp sequences were included in a phylogenetic analysis with one exception as the sequence was too short (185 bp). A Maximum-Likelihood (ML) analysis was performed with the 21 remaining sequences obtained in the study (GenBank accession numbers ON652986-ON653001 and OP727997-OP728001) and 69 reference AstV RdRp partial nucleo-tide sequences. The sequence alignment was obtained with Clustal Omega 1.2.3 (with initial and refinement iteration guide trees obtained by fast clustering; mBed algorithm). The ML analysis was performed with the program PhyML 3.1 [23], with the TVM model, an estimation of the proportion of invariable sites (I) and of the nucleotide heterogeneity of substitution rates (α), as selected by Model Generator 0.85 [24], and 1000 bootstraps.

Results

AstV RdRp was sequenced in 22 of the 515 samples tested (Table 1). Only one individual of the most abundant endemic species living outside of native forest areas tested positive (*Micro-gale brevicaudata*, family Tenrecidae, n = 98). That animal was captured in secondary forest. The remaining positive samples were from introduced rodents Muridae *Rattus rattus* (5.6%, 15/266) and *Mus musculus* (7.7%, 6/78). Given that these two species had the highest positivity rate, we focused our statistical analyses on them (n = 296). No significant difference in the infection frequency was detected between *Rattus* and *Mus* (Fisher's exact test, p = 0.59). Positivity rate was significantly higher in Manantenina in 2019 (12.1%, 17/140) than in Mandena in 2017 (0.9%, 1/112) or 2018 (1.5%, 3/259; p<0.001).

Positivity rates varied by species and habitat (see Fig 2 for introduced rodents). Overall, the highest positivity rate of AstVs infecting small mammals was in brushy regrowth (17.5%, 7/

Species	E / I	Semi-intact forest	Secondary forest	Brushy regrowth	Agroforest	Flooded rice	Village
Eliurus ellermani	Е	NA	0/1(0)	NA	NA	NA	NA
Eliurus webbi	Е	0/2(0)	0/3(0)	NA	NA	NA	NA
Nesomys audeberti	Е	0/2(0)	NA	NA	NA	NA	NA
Microgale brevicaudata	Е	0/31(0)	1/17(5.88)	0/5(0)	0/3(0)	0/42(0)	NA
Microgale longicaudata	Е	NA	NA	NA	0/1(0)	NA	NA
Setifer setosus	Е	0/5(0)	0/3(0)	0/2(0)	0/2(0)	0/6(0)	NA
Tenrec ecaudatus	Е	NA	NA	NA	NA	0/2(0)	NA
Rattus rattus	Ι	0/43(0)	2/43(4.65)	6/23(26.09)	1/14(7.14)	3/81(3.7)	3/62(4.84)
Mus musculus	Ι	NA	NA	1/9(11.11)	0/6(0)	5/22(22.73)	0/41(0)
Suncus etruscus	Ι	NA	0/5(0)	0/1(0)	NA	0/7(0)	NA
Suncus murinus	Ι	0/1(0)	0/2(0)	NA	0/2(0)	0/14(0)	0/12(0)

Table 1. Astrovirus detection in small terrestrial Malagasy mammals captured in and outside of Marojejy National Park. Species that were found positive (*i.e.* positive through hemi-nested PCR and with a Sanger sequence) are emphasized in bold. "E" refers to Endemic small mammals and "I" refers to Introduced small mammals.

https://doi.org/10.1371/journal.pntd.0012263.t001





https://doi.org/10.1371/journal.pntd.0012263.g002

40), and the lowest in flooded rice fields (4.6%, 8/174), secondary forest (4.1%, 3/74), agroforest (3.6%, 1/28), village (2.6%, 3/115), and semi-intact forest (0%, 0/84). These differences in overall positivity rates were statistically significant with a Fisher's exact test (p = 0.005; Fig 2). The frequency of infected individuals showed a trend to be slightly higher in brushy regrowth than in semi-intact forest (6 vs 0, pairwise Fisher's tests with Bonferroni correction, p = 0.062) and in brushy regrowth than in village settings (6 vs 3, p = 0.067).

The 95% cumulative sum Akaike weight subset of models contained 19 models. The highest weighted model (weight = 0.254) had only one parameter: village. The remaining models had ≤ 0.10 weights. The sum of weights of each predictor as determined by the sum of weights of each predictor in all models in descending order are village (weight = 1.00) across all 19 models, sex (weight = 0.26) across 8 models, season (weight = 0.24) across 7 models, mass (weight = 0.24) across 7 models, species (weight = 0.24) across 7 models, and habitat type (weight = 0.12) across 5 models. Despite their presence in the best model subset, the 95% CI of all of these estimates contained 1, thus none of these predictors significantly impacted the odds of infection, with the notable exception of "village". Full model-averaged parameter estimates are provided in Fig 3. The complete table of all model combinations and coefficients is provided in S1 Table.

A second model with only the data from the village of Manantenina was considered because positivity rate was significantly higher there (n = 102). The full model investigated the effect of four variables: habitat type (secondary forest, agroforest, flooded rice, village), species (*Rattus rattus, Mus musculus*), sex, and mass. Season was not considered because trapping only occurred in the dry season. Across the 16 models, the intercept only model had the highest Akaike weight (0.21), and the remaining models had ≤ 0.16 weights. The sum of weights of each predictor in descending order were habitat type (weight = 0.37) across 5 models, sex (weight = 0.32) across 5 models, mass (weight = 0.25) across 5 models, and species (weight = 0.22) across 4 models. Full model-averaged parameter estimates are provided in S2 Table.



Fig 3. Model-averaged results of logistic regression predicting AstV infection based on habitat type (with semiintact forest as the reference level), species (with *Rattus* as the reference level), sex (with female as reference level), and body mass.

https://doi.org/10.1371/journal.pntd.0012263.g003

We conducted a phylogenetic analysis of the AstVs that were detected in these small mammals. The overall sequence pairwise similarity among the 21 obtained sequences ranged from 46% to 100% and clustered in three phylogenetic groups (Fig 4). The first group included sequences obtained from AstVs found in *Rattus rattus* (n = 14) and the one positive *Microgale brevicaudata*. The second group contained only sequences from *Mus musculus* (n = 6), clustering with an AstV sequence reported from Afghanistan. The remaining AstV sample detected in *Rattus rattus* clustered at the root of the avian astrovirus clade (i.e., the Avastrovirus clade). Of note, none of the sequences obtained in this study were related to those obtained from AstVs detected in Malagasy bats (Fig 4).

Discussion

Although there is no information regarding the exposure of the Malagasy population to AstVs, presented data suggest that human populations they may be exposed to those viruses, at least in rural settings. The screening of 515 small terrestrial mammals revealed that Astrovirus infection is almost exclusively limited to introduced rodents, which typically are more common in habitats disturbed by human activities, as compared to native forest formations [25]. We found only a single individual of an endemic species infected with AstVs, *Microgale brevicaudata*, which is one of the few endemic species able to adapt to human disturbed habitats in this area of Madagascar [17,26]. However, the relatively small number of samples of endemic Nesomyinae rodents and Tenrecidae tenrecs limits our ability to assess the importance of these mammals in AstV maintenance at the study site. It is also noteworthy that introduced rodents exhibited AstV infection but not introduced shrews (genus *Suncus*, Table 1), in contrast with a recent investigation reporting 7.4% of AstVs infection in *Suncus murinus* in Singapore [9].

The phylogenetic tree of the AstVs is structured by host species, with one strongly supported clade composed of sequences from mice, a second supported clade composed of viral sequences obtained from rats with the notable exception of an AstV sequence from endemic *Microgale brevicaudata*, and a last sequence obtained from a rat and clustering in the Avastrovirus clade, which typically infects birds. Of note, no birds were sampled at the study site and the obtained sequence was distinct from any sequence available in databases, so the possibility of a contamination must be considered as low. The topology of the phylogenetic tree and infection almost exclusively limited to introduced rodents support that these AstVs were introduced to Madagascar with murid rodents and with limited spillover to endemic species. Current archaeological evidence indicates that *Mus* was introduced to the island during the



Fig 4. Midpoint rooting maximum likelihood tree obtained with 90 Astrovirus (AstV) RNA-dependent RNApolymerase partial nucleotide sequences (389 bp). The phylogenetic analysis was conducted with the TVM + I + Γ evolutionary model (I = 0.14; α = 1.04) and 1000 bootstraps. Sequences derived from this study are colored in red. AstVs previously detected in Malagasy bats are colored in orange. Bootstrap values are reported when higher than 80. Scale bar: mean number of nucleotide substitutions per site.

https://doi.org/10.1371/journal.pntd.0012263.g004

11th century via Islamic trade routes [27] while the introduction of *Rattus rattus* on Madagascar is thought to have originated from India or the Middle East [28,29]. Further investigation on the origin of rodent AstVs in Madagascar could be assessed by sampling additional rodents in Madagascar and in the Middle East, along with more detailed studies of virus circulation in the islands of the western Indian Ocean and neighboring East Africa. Major differences between rodent and bat AstVs also suggest that these viruses arrived on Madagascar by different dispersal events, with the bat viruses via migration from continental areas and the rodent viruses via human introductions.

On Madagascar, Rattus rattus is highly adaptable, found in diverse habitats across the sampled gradient from semi-intact forest to secondary growth and agricultural fields, and in village homes [30]. It is the most frequently captured rodent in many field surveys on the island, including those in forests transformed by human activities [31]. Although introduced rodents were present in all sampled habitats, some results indicated that habitat type influences positivity rate, with the lowest positivity rate found in semi-intact or secondary forests. This pattern may result from small mammal community structure sheltered by these less disturbed habitats: introduced rodents exist at lower abundance in our sampled forest habitats, which could lower transmission among competent reservoirs. The majority of cases were found in the brushy regrowth (n = 6) and flooded rice fields (n = 8), with the latter sheltering an abundant *Mus* population displaying the highest positivity rate in our study. Despite seeing significant differences between species and habitat types using pairwise comparisons we were unable to determine what demographic/morphometric indices (species, age, sex, mass) might be driving these differences using a GLM. However, given the number of covariates in the models and the low positivity rate, only a strong effect would have been significant in our models. Several studies have highlighted the ability of AstVs to remain infectious for several weeks in surface water and especially in groundwater due to favorable temperature, limited exposure to bacterial agents [32–35]. Such a high persistence ability may favor transmission among Mus of AstVs in rice fields. However, this pattern was not observed in Rattus, and hence deserves cautious interpretation. Furthermore, we found that the village in which animals were trapped was the most important predictor of AstV infections, with positivity rate being substantially higher in Manantenina (12.1%, 17/140) than Mandena (1.3%, 5/375, p<0.001 fisher's exact test). In Manantenina, habitat type was a weak predictor of infection. Because sampling was conducted asynchronously between villages, we are unable to determine if this effect is due to a true difference between the villages or varying annual positivity rate of AstVs.

The detection of a "*Rattus*" AstV in *Microgale brevicaudata* and of an avian AstV in *Rattus rattus* highlights the potential for host shifting of these viruses [36]. This study and a previous investigation on bat AstV [14] provides an initial assessment of the diversity of AstV associated with Malagasy mammals, but additional research is needed to investigate AstVs infection in wild avian hosts, as well as in avian and mammalian livestock and humans, to fully assess AstV spillover in natural and human-modified habitats.

Our study has some obvious limitations, including the randomness of the sampling strategy. Although the trapping design aimed at trapping most of the small mammal diversity, some species might have been under sampled, especially endemics. Our study thus does not allow addressing the importance of endemic rodents in the maintenance of AstVs because of the low number of sampled animals and because these endemic rodents might shelter AstV that are not detected by our screening PCR scheme. However, these species have few contacts with humans as they mostly occupying intact or semi intact forests. Altogether, presented data have implications for the potential transmission of zoonotic pathogens from introduced rodents to people, especially in degraded habitats, although to our knowledge there is no available data on AstV epidemiology in human populations from Indian Ocean islands. Firstly, the topology of the presented phylogenetic tree strongly suggests that AstVs infecting introduced rodents at the study site were introduced to Madagascar, likely from Asia, although complementary studies such as the investigation of AstVs on the other different islands of the region are necessary to confirm this. Secondly, our findings show that human exposure increases in disturbed habitats due to (i) a higher AstV positivity rate in animal reservoirs within these habitats, as well as (ii) an increased abundance of invasive rodent hosts. The land-use types in which people spend the most time may present the greatest risk of exposure to AstVs from terrestrial small mammals, which in turn has implications for diarrheal diseases in rice farmers. Altogether, our data encourage the need for assessing the actual exposure of the local population, especially farmers, to the AstVs described herein.

Supporting information

S1 Table. Model combinations and coefficients. (XLSX)

S2 Table. Full model-averaged parameter estimates. (CSV)

Acknowledgments

Hugo Tavernier assisted FB for the screening of the samples during an undergraduate internship.

Author Contributions

- **Conceptualization:** Steven M. Goodman, Charles L. Nunn, Camille Lebarbenchon, Pablo Tortosa.
- Data curation: James P. Herrera, Kayla Kaufman, Camille Lebarbenchon, Pablo Tortosa.
- Formal analysis: Victoria Carcauzon, James P. Herrera, Kayla Kaufman, Charles L. Nunn, Camille Lebarbenchon, Pablo Tortosa.

Funding acquisition: Charles L. Nunn, Pablo Tortosa.

- **Investigation:** Victoria Carcauzon, James P. Herrera, Kayla Kaufman, Fiona Baudino, Natalie Wickenkamp, Toky M. Randriamoria, Voahangy Soarimalala, Steven M. Goodman, Camille Lebarbenchon, Pablo Tortosa.
- Methodology: Kayla Kaufman, Voahangy Soarimalala, Steven M. Goodman, Charles L. Nunn, Camille Lebarbenchon, Pablo Tortosa.
- Project administration: Charles L. Nunn.

Supervision: Charles L. Nunn, Pablo Tortosa.

Validation: Camille Lebarbenchon, Pablo Tortosa.

Visualization: Kayla Kaufman, Camille Lebarbenchon.

Writing – original draft: Victoria Carcauzon, James P. Herrera, Kayla Kaufman, Fiona Baudino, Natalie Wickenkamp, Toky M. Randriamoria, Voahangy Soarimalala, Steven M. Goodman, Charles L. Nunn, Camille Lebarbenchon, Pablo Tortosa. Writing – review & editing: Victoria Carcauzon, James P. Herrera, Kayla Kaufman, Fiona Baudino, Natalie Wickenkamp, Toky M. Randriamoria, Voahangy Soarimalala, Steven M. Goodman, Charles L. Nunn, Camille Lebarbenchon, Pablo Tortosa.

References

- Gaggero AO'Ryan M, Noel JS, Glass RI, Monroe SS, Mamani N, et al. Prevalence of astrovirus infection among Chilean children with acute gastroenteritis. J Clin Microbiol. 1998; 36: 3691–3693. <u>https://doi.org/10.1128/JCM.36.12.3691-3693.1998</u> PMID: 9817899
- Boga JA, Melón S, Nicieza I, I de Diego, Villar M, Parra F, et al. Etiology of sporadic cases of pediatric acute gastroenteritis in Asturias, Spain, and genotyping and characterization of norovirus strains involved. J Clin Microbiol. 2004; 42: 2668–2674. https://doi.org/10.1128/JCM.42.6.2668-2674.2004 PMID: 15184450
- Gosert R, Heininger U, Hirsch HH. Enterovirus detection in patients with acute gastroenteritis in Switzerland. J Med Virol. 2018; 90: 685–691. https://doi.org/10.1002/jmv.25005 PMID: 29236304
- Shastri S, Doane AM, Gonzales J, Upadhyayula U, Bass DM. Prevalence of astroviruses in a children's hospital. J Clin Microbiol. 1998; 36: 2571–2574. <u>https://doi.org/10.1128/JCM.36.9.2571-2574.1998</u> PMID: 9705394
- 5. Roach SN, Langlois RA. Intra and cross-species transmission of astroviruses. Viruses. 2021; 13: 1127. https://doi.org/10.3390/v13061127 PMID: 34208242
- Wunderli W, Meerbach A, Guengoer T, Berger C, Greiner O, Caduff R, et al. Astrovirus infection in hospitalized infants with severe combined immunodeficiency after allogeneic hematopoietic stem cell transplantation. PLOS ONE. 2011; 6: e27483. <u>https://doi.org/10.1371/journal.pone.0027483</u> PMID: 22096580
- 7. Vu D-L, Cordey S, Brito F, Kaiser L. Novel human astroviruses: Novel human diseases? J Clin Virol. 2016; 82: 56–63. https://doi.org/10.1016/j.jcv.2016.07.004 PMID: 27434149
- 8. Phan TG, Kapusinszky B, Wang C, Rose RK, Lipton HL, Delwart EL. The fecal viral flora of wild rodents. PLOS Pathog. 2011; 7: e1002218. https://doi.org/10.1371/journal.ppat.1002218 PMID: 21909269
- Neves ES, Mendenhall IH, Borthwick SA, Su YCF, Smith GJD. Genetic diversity and expanded host range of astroviruses detected in small mammal hosts in Singapore. One Health. 2021; 100218. https://doi.org/10.1016/j.onehlt.2021.100218 PMID: 33553564
- To KKW, Chan W-M, Li KSM, Lam CSF, Chen Z, Tse H, et al. High prevalence of four novel astrovirus genotype species identified from rodents in China. J Gen Virol. 2017; 98: 1004–1015. <u>https://doi.org/10. 1099/jgv.0.000766</u> PMID: 28537544
- Mombo IM, Suquet E, Boundenga L, Mveang-Nzoghe A, Maganga-Mboga C, Arnathau C, et al. Detection of novel astroviruses among rodents of Gabon, Central Africa. Infect Genet Evol J Mol Epidemiol Evol Genet Infect Dis. 2019; 68: 43–46. https://doi.org/10.1016/j.meegid.2018.12.003 PMID: 30529088
- Fischer K, Pinho dos Reis V, Balkema-Buschmann A. Bat astroviruses: Towards understanding the transmission dynamics of a neglected virus family. Viruses. 2017; 9: 34. <u>https://doi.org/10.3390/ v9020034 PMID: 28230787</u>
- 13. Hoarau F, Le Minter G, Joffrin L, Schoeman MC, Lagadec E, Ramasindrazana B, et al. Bat astrovirus in Mozambique. Virol J. 2018; 15: 104. https://doi.org/10.1186/s12985-018-1011-x PMID: 29925396
- Lebarbenchon C, Ramasindrazana B, Joffrin L, Bos S, Lagadec E, Le Minter G, et al. Astroviruses in bats, Madagascar. Emerg Microbes Infect. 2017; 6: e58. <u>https://doi.org/10.1038/emi.2017.47</u> PMID: 28634357
- Joffrin L, Hoarau AOG, Lagadec E, Köster M, Ramanantsalama RV, Mavingui P, et al. Astrovirus in Reunion free-tailed bat (*Mormopterus francoismoutoui*). Viruses. 2021; 13: 1524. <u>https://doi.org/10.3390/v13081524 PMID: 34452389</u>
- 16. Goodman S, Soarimalala V, Olson L. Systematics of endemic Malagasy tenrecs (family Tenrecidae). The terrestrial protected areas of Madagascar: their history, description, and biota. Antananarivo: Association Vahatra; 2018. pp. 363–372.
- Herrera JP, Wickenkamp NR, Turpin M, Baudino F, Tortosa P, Goodman SM, et al. Effects of land use, habitat characteristics, and small mammal community composition on *Leptospira* prevalence in northeast Madagascar. PLoS Negl Trop Dis. 2020;14. https://doi.org/10.1371/journal.pntd.0008946 PMID: 33382723
- Seltmann A, Corman VM, Rasche A, Drosten C, Czirják GÁ, Bernard H, et al. Seasonal fluctuations of astrovirus, but not coronavirus shedding in bats inhabiting human-modified tropical forests. Ecohealth. 2017; 14: 272–284. https://doi.org/10.1007/s10393-017-1245-x PMID: 28500421

- Cotton S, Hending D, Andrianiaina A. Biodiversity assessment of vanilla plantations and surrounding areas in the SAVA region of Madagascar. Open Science Centre, University of Jyväskylä; 2018. https:// doi.org/10.17011/conference/eccb2018/107496
- 20. Goodman SM, Raherilalao MJ, Wohlhauser S, editors. Les aires protégées terrestres de Madagascar: leur histoire, description et biota, tome 1: Introduction. Antananarivo, Associations Vahatra; 2021.
- Soarimalala V, Goodman S. Les petits mammifères de Madagascar. Antananarivo, Association Vahatra; 2011.
- 22. Chu DKW, Poon LLM, Guan Y, Peiris JSM. Novel astroviruses in insectivorous bats. J Virol. 2008; 82: 9107–9114. https://doi.org/10.1128/JVI.00857-08 PMID: 18550669
- Guindon S, Dufayard J-F, Lefort V, Anisimova M, Hordijk W, Gascuel O. New algorithms and methods to estimate maximum-likelihood phylogenies: assessing the performance of PhyML 3.0. Syst Biol. 2010; 59: 307–21. https://doi.org/10.1093/sysbio/syg010 PMID: 20525638
- Keane TM, Creevey CJ, Pentony MM, Naughton TJ, McInerney JO. Assessment of methods for amino acid matrix selection and their use on empirical data shows that ad hoc assumptions for choice of matrix are not justified. BMC Evol Biol. 2006; 6: 29. https://doi.org/10.1186/1471-2148-6-29 PMID: 16563161
- Allnutt TF, Asner GP, Golden CD, Powell GVN. Mapping recent deforestation and forest disturbance in northeastern Madagascar. Trop Conserv Sci. 2013; 6: 1–15. <u>https://doi.org/10.1177/</u> 194008291300600101
- Rasolobera F, Randriamoria TM, Soarimalala V, Goodman S. Écologie de la reproduction de Microgale brevicaudata (Tenrecidae) dans le Nord-Est de Madagascar. Bull Soc Zool Fr. 2022; 147: 143–151.
- Duplantier J-M, Orth A, Catalan J, Bonhomme F. Evidence for a mitochondrial lineage originating from the Arabian peninsula in the Madagascar house mouse (*Mus musculus*). Heredity. 2002; 89: 154–158. https://doi.org/10.1038/sj.hdy.6800122 PMID: 12136419
- Tollenaere C, Brouat C, Duplantier J-M, Rahalison L, Rahelinirina S, Pascal M, et al. Phylogeography of the introduced species *Rattus rattus* in the western Indian Ocean, with special emphasis on the colonization history of Madagascar. J Biogeogr. 2010; 37: 398–410. <u>https://doi.org/10.1111/j.1365-2699</u>. 2009.02228.x
- Yu H, Jamieson A, Hulme-Beaman A, Conroy CJ, Knight B, Speller C, et al. Palaeogenomic analysis of black rat (*Rattus rattus*) reveals multiple European introductions associated with human economic history. Nat Commun. 2022; 13: 2399. https://doi.org/10.1038/s41467-022-30009-z PMID: 35504912
- Ramasindrazana B, Randriamoria T, Rahelinirina S, Duchemin JB, Goodman S. Introduced terrestrial small mammals. Princeton University Press. In The New Natural History of Madagascar. Princeton University Press. 2022. pp. 1872–1880.
- Duplantier J-M, Rakotondravony D. The rodent problem in Madagascar: Agricultural pest and threat to human health. Australian Centre for International Agricultural Research. In Ecologically-based Management of Rodent Pests. Australian Centre for International Agricultural Research. Canberra; 1999. pp. 441–459.
- Abad FX, Pintó RM, Villena C, Gajardo R, Bosch A. Astrovirus survival in drinking water. Appl Environ Microbiol. 1997; 63: 3119–3122. https://doi.org/10.1128/aem.63.8.3119-3122.1997 PMID: 9251198
- 33. Bosch A, Pintó RM, Guix S. Human astroviruses. Clin Microbiol Rev. 2014; 27: 1048–1074. <u>https://doi.org/10.1128/CMR.00013-14 PMID: 25278582</u>
- Espinosa AC, Mazari-Hiriart M, Espinosa R, Maruri-Avidal L, Méndez E, Arias CF. Infectivity and genome persistence of rotavirus and astrovirus in groundwater and surface water. Water Res. 2008; 42: 2618–2628. https://doi.org/10.1016/j.watres.2008.01.018 PMID: 18291437
- Pintó RM, Villena C, Le Guyader F, Guix S, Caballero S, Pommepuy M, et al. Astrovirus detection in wastewater samples. Water Sci Technol J Int Assoc Water Pollut Res. 2001; 43: 73–76. PMID: 11464772
- Mendenhall IH, Smith GJD, Vijaykrishna D. Ecological drivers of virus evolution: astrovirus as a case study. J Virol. 2015; 89: 6978–6981. https://doi.org/10.1128/JVI.02971-14 PMID: 25948751